Author’s response to reviews

Title: Three Intellectual Disability-associated De novo Mutations in MECP2 Identified by Trio- WES Analysis

Authors:

Yi Gu (kitty-84@163.com)
Bingwe Xiang (15058302377@163.com)
Lina Zhu (zhulina507@126.com)
Xiuwei Ma (ponyo07@vip.sina.com)
Xiang Chen (chenxiangnj2005@aliyun.com)
Tao Cai (tcai@mail.nih.gov)

Version: 2 Date: 30 Apr 2020

Author’s response to reviews:

Dear Dr. Arwa Tawfiq,

Thank you so much for your quick response, careful review, and kind suggestions -- we really appreciate!
We have revised all the places as you pointed out, and have made the data public in Mendeley Data.
Thank you for your consideration.
With best regards,
Tao
NIH
tcai@mail.nih.gov
301-402-5320

Assistant Editor Comments:

1. Corresponding author email

Please list only one email for the corresponding author in the submission system, ensuring that this matches the email listed on the title page of your manuscript (currently tcai@mail.nih.gov).

Response: I have updated my information in the submission system. The email has been unified as tcai@mail.nih.gov.

2. Ethics approval and consent to participate

Please remove the following statement from the Ethics approval and consent to participate section of your manuscript:
As a disclaimer, T.C. represented his own perspective in the article, not that of the National Institute of Dental and Craniofacial Research or the National Institutes of Health.

Response: This statement has been removed in “Ethics approval and consent to participate”.

3. Consent for publication

Please remove the following statement from the Consent for publication section of your manuscript: “This study was authorized by Ethical Committee of Wenzhou Medical University.”

Response: This statement has been removed in “Consent for publication”.

4. Availability of data and materials

a) Please note that all data must be made public prior to publication and we cannot proceed to publish your manuscript if your data is not deposited and made publicly available. We are an open access journal and submission of a manuscript to a BMC journal implies that materials described in the manuscript, including all relevant raw data, will be freely available to any scientist wishing to use them for non-commercial purposes. By raw data we mean the minimal dataset that would be necessary to interpret, replicate and build upon the findings reported in the article.

As your data falls under the list of data types which must be deposited (listed below), we ask that you please provide the relevant accession numbers, and the name of the database in which your data is stored, if your data has been deposited into a database or the appropriate web links if the data has been uploaded into a repository. If the data has not yet been deposited, please do so and provide the relevant information needed to access it in the ‘Availability of data and materials’ section in your manuscript. Please ensure all data is already released and that any accession numbers/web links are in the ‘Availability of Data and Materials’ in their final form, and that all web links and accessions allow access to public data.

Your data availability statement can take the following form: “The datasets generated and/or analysed during the current study are available in the [NAME] repository, [PERSISTENT WEB LINK OR ACCESSION NUMBER TO DATASETS]”.

Response: We have made the data public in Mendeley Data, and add statement as you suggested in “Availability of data and materials”.

b) Please ensure that you include the accession numbers or direct links and the full names of the data banks/repositories corresponding to any datasets obtained from web-based sources and subsequently analysed in your study in the Availability of Data and Materials section, including any reference datasets. If accession numbers or web links are listed in a table or additional file, please include an in-text reference to the file or table in the Availability of Data and Materials section.

Response: We have included the accession number of “doi:10.17632/34v7c9rkgf.1” and the full name “Mendeley Data” of the data repositories in the “Availability of data and materials”.

5. Competing interests

Please amend the following statement in the Competing interests section of your manuscript, from: “As author T.C. is a member of the editorial board (Associate Editor) of this journal.”  to:
“The author Tao Cai is a member of the editorial board for the BMC Medical Genetics journal”.

Response: This statement has been amended as you suggested in “Competing interests”.

6. Authors’ contributions

Please consider the list of authors as it currently stands with reference to our guidelines regarding qualification for authorship (http://www.biomedcentral.com/submissions/editorial-policies#authorship).

Currently, the contributions of authors BX, LZ, XM, and XC does not automatically qualify them for authorship. In the section “Authors’ contributions”, please provide further clarifications on their contributions, and see our guidelines for authorship below.

An 'author' is generally considered to be someone who has made substantive intellectual contributions to a published study. Authors are expected to fulfil the criteria below (adapted from McNutt et al., Proceedings of the National Academy of Sciences, Feb 2018, 201715374; DOI: 10.1073/pnas.1715374115; licensed under CC BY 4.0):

Each author is expected to have made substantial contributions to the conception OR design of the work; OR the acquisition, analysis, OR interpretation of data; OR the creation of new software used in the work; OR have drafted the work or substantively revised it

AND to have approved the submitted version (and any substantially modified version that involves the author's contribution to the study);

AND to have agreed both to be personally accountable for the author's own contributions and to ensure that questions related to the accuracy or integrity of any part of the work, even ones in which the author was not personally involved, are appropriately investigated, resolved, and the resolution documented in the literature.

Please note that acquisition of funding, collection of data or general supervision of the research group, alone, does not usually justify authorship. You may acknowledge individuals who assist in the acquisition of funding or of data or supervise the research group in the Acknowledgments section of your manuscript.

If you wish to amend the list of authors, please use the standardised form which you and your co-authors must complete. The authorship change form can be found at the following link: https://resource-cms.springernature.com/springer-cms/rest/v1/content/7454878/data/v5

All instructions can be found on the form, please treat the 'current authorship' section as the original authorship. Please return the form within 14 days by email to the Editorial Office with all author signatures (including those newly added/removed).

Response: Thank you for your reminding. We have checked and amended the statement in “Authors’ contributions” as you suggested. We agree with the current list of authors. Thank you for your kind help.

7. Remove attachments

Please remove the following additional files from your manuscript and from the file inventory as they
are no longer needed at this stage:
• Response to editor and 2 reviewers for MGTC-D-19-00394.docx
• Response to editor for MGTC-D-20-00197R1.docx

Response: These two attachments have been removed.

8. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Should you wish to respond to these revision requests, please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter.

Response: We have checked the manuscript and submit the responses as you suggested.